Author’s response to reviews

Title: Risk factors for renal toxicity after inpatient cisplatin administration

Authors:
Elena Galfetti (elena.galfetti@ksw.ch)
Alessandra Cerutti (studiomedicocerutti@gmail.com)
Michele Ghielmini (michele.ghielmini@eoc.ch)
Emanuele Zucca (emanuele.zucca@eoc.ch)
Luciano Wannesson (luciano.wannesson@eoc.ch)

Version: 2 Date: 20 Feb 2020

Author’s response to reviews:

Dear Editor,

I would like to thank you very much again for having accepted the opportunity to check with our local ethical committee the correctness of the ethical aspects of our original paper.

I am happy to inform that the Ethical Committee decided in their session of February 18th 2020 that the research article “Risk factors for renal toxicity after inpatient cisplatin administration” that I am submitting to BMC Pharmacology and Toxicology do not have any ethical flaws and can be published as is.

Contact information:
Comitato etico cantonale
c/o Ufficio di sanità
Via Orico 5
6501 Bellinzona (Switzerland)

Tel. +41 91 814 30 57 / Email address: dss-ce@ti.ch / Web page: https://www4.ti.ch/dss/dsp/us/ce/comitato/